Author's response to reviews

Title: Case report: Infective endocarditis caused by Brevundimonas vesicularis

Authors:

Mei-Li Yang (940424@ms.kmuh.org.tw)
Yen-Hsu Chen (d810070@cc.kmu.edu.tw)
Wei-Ru Lin (lwru@mail2000.com.tw)
Tun-Chieh Chen (dalechen.chen@gmail.com)
Chun-Yu Lin (infectionman@gmail.com)
Po-Liang Lu (d830166@cc.kmu.edu.tw)

Version: 5 Date: 15 December 2006

Author’s response to reviews:

Dec. 17, 2006
Annabel Phillips PhD
Assistant Editor
BMC-series Journals
BioMed Central
Middlesex House
34-42 Cleveland Street
London W1T 4LB

Tel: +44 (0)20 7631 9138
Fax: +44 (0)20 7636 4296 / +44 (0)20 7631 9923
email: editorial@biomedcentral.com
Web: www.biomedcentral.com

Dear Editors,
Regarding to 2095928401146748 - Case report: Infective endocarditis caused by Brevundimonas vesicularis, we would like to have changes the following according to:
1) Within the background section of the abstract, please include more detail to provide the context and purpose of the case.
The the background section of the abstract will be revised according to the suggestion of editors.
2) Include a 'competing interests' section -
We make our declaration of no competing interests in the revised article.

With best regards
Po-Liang Lu, M.D.

Po-Liang Lu,
Department of Internal Medicine,
Kaohsiung Medical University Hospital,
No. 100, Tzyou 1st Road 807, Kaohsiung, Taiwan
Telephone: 886-7-3121101 ext 5675
FAX: 886-7-3228547
E-mail: d830166@cc.kmu.edu.tw